Author’s response to reviews

Title: Food choice patterns of long-haul truck drivers driving through Germany, a cross sectional study.

Authors:
Andreas Bschen (andreas.bschen@uni-hohenheim.de)
Siegfried Rothe (Siegfried.Rothe@daimler.com)
Anja Schöner (anja.schoener@web.de)
Nina Pijahn (ninapijahn@googlemail.com)
Nanette Stroebele-Benschop (n.stroebele@uni-hohenheim.de)

Version: 4 Date: 13 Nov 2019

Author’s response to reviews:

Dear Mr. Madsen,

we have picked a data storage that requires signing up for on purpose, however, given your concerns we have replaced the link with the following statement: "The datasets used and analysed during the current study are available from the corresponding author on reasonable request".

Thank you for considering the manuscript for publication in BMC Nutrition.

Sincerely,

Nanette Stroebele-Benschop